Author's response to reviews

Title: Previous experiences and emotional baggage as influences on lifestyle change - a qualitative study of Norwegian Healthy Life Centre participants

Authors:

Ingrid S Følling (ingfoll@gmail.com)
Marit Solbjør (marit.solbjor@svt.ntnu.no)
Anne S Helvik (anne-sofie.helvik@ntnu.no)

Version: 2 Date: 8 January 2015

Author's response to reviews: see over
Dear Executive Editor

Please find enclosed the manuscript “Previous experiences and emotional baggage as influences on lifestyle change - a qualitative study of Norwegian Healthy Life Centre participants” which we hope you will consider for publication in the BioMed Central Public Health.

This research was funded by North-Trondelag University College.

The article has not been published or submitted elsewhere. The regional Committee for Medical Research Ethics in Middle Norway approved the study. There is no area of conflict in this article.

The undersigned authors have contributed to the conception, conduct and analyses of the study and preparation of the manuscript according to the requirements of your journal.

Sincerely yours,

Ingrid S. Følling
Marit Solbjør
Anne-Sofie Helvik

Corresponding author
Ingrid S. Følling
Department of Public Health and General Practice
Faculty of Medicine
Norwegian University of Science and Technology
7491 Trondheim
Norway
Tlf: +47 74 02 25 97/ +47 92 85 54 12
E-mail: ingfoll@gmail.com